Author’s response to reviews

Title: Papillary Thyroid Carcinoma Presented as a Primary Renal Tumor with Multiple Pulmonary and Bone Metastases: A Case Report

Authors:

Emre Gezer (gezer_emre@hotmail.com)
Alev Selek (alevselek@gmail.com)
İlhan Tarkun (ilhantarkun@superonline.com)
Zeynep Cantürk (zeynepcanturk@hotmail.com)
Berrin Çetinarslan (barslan@kocaeli.edu.tr)

Version: 2 Date: 18 Feb 2019

Author’s response to reviews:

Additional paragraph at the end of the background part that explains why this case report is presented is added.

The past medical, family, employment, prior medications including their doses and social (smoking and consuming alcohol) histories were added.

Physical examination including blood pressure and the pulse was added. However, neurological examination and her temperature on admission couldn’t be discovered on her patient folder. Laboratory findings on admission were given.

The medical details after radiotherapy cannot be specified due to her lost to follow-up, but it was described how we got the information about the patient.

In discussion part, the 2nd paragraph in which describing the case and our aim presenting it was replaced with the 1st paragraph.